Author's response to reviews

Title: Do physician outcome judgments and judgment biases contribute to inappropriate use of treatments?
Study Protocol

Authors:

Jamie C Brehaut (jbrehaut@ohri.ca)
Roy Poses (rposes@firmfound.org)
Kaveh G Shojania (kshojania@ohri.ca)
Alison Lott (alott@ohri.ca)
Malcolm Man-Son-Hing (mhing@ohri.ca)
Elise Bassin (elise_bassin@post.harvard.edu)
Jeremy Grimshaw (jgrimshaw@ohri.ca)

Version: 5 Date: 25 May 2007

Author's response to reviews:

The attached final draft includes the following changes:

- 'Abbreviated terms' was changed to 'Abbreviations'
- 'Author's Contributions' was changed to 'Authors' Contributions'
- It was indicated under the 'Authors' Contributions' heading that all authors read and approved the final manuscript
- A sentence was added under the 'Competing Interests' section stating that the authors do not have any competing interests.
- The additional files were removed
- The reference list was changed to include the names of all authors (i.e. 'et al' was removed)

Thank you,
JCB